Author's response to reviews

Title: The Prevalence, Risk Factors, and Screening Measure for Prediabetes and Diabetes among Emirati Overweight/Obese Children and Adolescents

Authors:

Elham Al Amiri (elham.amiri@moh.gov.ae)
Mona Abdullatif (mona.abdullatif@gmail.com)
Abdishakur Abdulle (asmabdulle@gmail.com)
Nibal Al Bitar (nmf-albitar@hotmail.com)
Elham Zaki Afandi (elham_afandi@live.com)
Monira Parish (drmp_82@yahoo.com)
Gassan Darwiche (ghassan.darwiche@med.lu.se)

Version: 2 Date: 24 May 2015

Author's response to reviews:

Please note that the following changes has been done as required:

1.) Email address of all authors in the Title page has been added.
2.) Line Numbering has been inserted.
3.) A List of abbreviations has been provided.
4.) The funding statement has been placed in the Acknowledgements section.